Author's response to reviews

Title: A new non-invasive approach based on polyhexamethylene biguanide increases the regression rate of HPV infection

Authors:

Antonio Gentile (gentileantonio.me@gmail.com)
Sandro Gerli (gerber@unipg.it)
Gian Carlo Di Renzo (direnzo@unipg.it)

Version: 2 Date: 13 April 2012

Author's response to reviews:

Dear Editor,

Further to your e-mail, all the changes required were made:

1) The trial registration number has been added to the end of the abstract
2) The sections "Competing interests", "Authors' contributions" and "Acknowledgements" have been added to the manuscript

Kindest regards,

Antonio Gentile